University of Toronto Department of Computer Science Summer 1997

CSC 148H: Introduction to Computer Science

Instructor

Philip Edmonds office: SF4008C phone: 978-5182 e-mail: pedmonds@cs.toronto.edu office hours: Mondays – Thursdays, 4:00–5:00 p.m., or by appointment

Lectures and Tutorials

Lectures: Wednesdays, 7:00–9:00 p.m., in MP102. Tutorials: Wednesdays, 6:00–7:00 p.m., see table for location.

Your birth month Tutor Room Jan–Apr Charles Chan MP102 May–Aug Qin Liu MP118^y Sep–Dec Piotr Sliz LM123

yIn SS1073 on May 28 only.

Tutors will hold lab hours at the Sigmund Samuel CDF-PC site. The dates and times will be posted on the course web page.

Course Materials

Read section 5 of the Course Handbook for a complete description of the required and optional items. You will probably buy (from the bookstore):

ffl Data Structures: An Object-Oriented Approach. Preliminary Edition. D.T. Barnard, R.C.

Holt, J.N.P. Hume. Holt Software Associates, Inc. 1995.

ffl Computer Science 148 Course Handbook. Fall 1996 edition. D. Horton (ed.).

ffl Computer Science 148 Lecture Notes. Fall 1996 edition. F. Fich and D. Horton.

Important: The Course Handbook contains general information about the course. The following amendments for the summer hold. There is no course coordinator in the summer, so all of your dealings will be with your instructor. The late policy is slightly changed to extend the late assignment deadline, as noted below. Section 6.1 erroneously says that your computer account name will have the form a148xxxx. Actually, it will have the form h148xxxx. See Appendix E about using your CDF-PC account.

Course Notices

Course web page: http://www.cs.toronto.edu/~pedmonds/148/summer97/

Handouts, assignments, solutions, marks, and important announcements will be posted on the web page. You should check it regularly, and especially before you hand in an assignment.

Grading Scheme

Assignments 40% Tests 20% Final Exam 40% 100%

To pass the course, you need a total mark of at least 50%, and you must receive at least 40% on both the exam and the term work (assignments plus tests).

Assignment Worth Due A1 – Warmup 5% week 3 A2 – ADTs 10% week 6 A3 – Pointers 10% week 9 A4 – Recursion warmup 1% week 10 A5 – Recursion; OOP 14% week 13

Test Worth When Quiz 1 5% week 4 Midterm 10% week 7 Quiz 2 5% week 11

Tests

The two half-hour quizzes will be held at the beginning of the tutorials in weeks 4 and 11. The midterm is one hour long, and will be held in the tutorial of week 7 (June 25). The exam will be three hours long, and will be held during the normal exam period of August 11–15.

Assignments – Late Policy

Assignments are due at 6 p.m. on Thursday. Assignments must be submitted to the 148 drop boxes on the second floor of the Sandford Fleming building, near room 2305A. (They are between the elevators and the bridge to the D.L. Pratt Building.)

The late policy is strict. You may submit an assignment late at most once during the term. The penalty for late assignments is 30%. For instance, an assignment that would have been worth 80/100 if it were on time, will earn 50/100 if submitted late. The deadline for late assignments is 9 a.m. on the Monday following the due date. NOTE: this replaces the deadline stated in

the Course Handbook. Consult the Course Handbook for other details.

You may request special consideration in the event of an illness or other catastrophe. Supporting documentation is required (e.g., a medical certificate). Fill out the ‘Request for Special Consideration’ form and return it and the documentation to your instructor.

Important Dates

Wednesday, June 11: Last day to switch to CSC108.

Friday, July 18: Last day to drop the course without academic penalty. Wednesday, August 6: Last lecture. August 11–15: Exam period.